Author's response to reviews

Title: Anomalous left coronary artery from the pulmonary artery (ALCAPA) presenting with sudden death in an octogenarian: A case report

Authors:

ahmad Separham (a.separham2005@yahoo.com)
Parvaneh Aliakbarzadeh (Aliakbarzadehp@yahoo.com)

Version: 7 Date: 21 October 2011

Author's response to reviews: see over
Anomalous left coronary artery from the pulmonary artery (ALCAPA) presenting with sudden death in an octogenarian: A case report

*Ahmad Separham1, Parvaneh Aliakbarzadeh2

Address1: Department of cardiology, Cardiovascular Research center, SHAHID MADANI HEART Hospital, Tabriz, Iran
Address2: Department of cardiology, Cardiovascular Research center, SHAHID MADANI HEART Hospital, Tabriz, Iran
E-mail1:a_separham2005@yahoo.com
E-mail2:Aliakbarzadehp@yahoo.com

Corresponding Author: Ahmad separham, Department of cardiology, Cardiovascular Research center, SHAHID MADANI HEART hospital, Tabriz, Iran

This is an interesting case of coronary anomaly because it may be the oldest case of this type anomaly and very rare cause of sudden death in such old age person.

All editors wanted points have been included in revised manuscript:
1. A sentence has been added to explain selective catheterization of RCA.
2. It has been noted that there is no clear cause-effect relationship between ALCAPA and VF in this old patient.
3. There is no evidence of her previous LVEF (this has been added to revised manuscript)
4. Language mistakes have been corrected: prolonged resuscitation has been converted to prolong resuscitation. Hemodynamically, led a sedentary life, hypokinetic have been deleted and alternative sentences have been added.
5. MR means mitral regurgitation.
6. discharged home in stable situation has been converted to another sentence.